Author's response to reviews

Title: Breast and spleen metastasis of squamous carcinoma from the uterine cervix: a case report

Authors:

meryem MA aitelhaj (m_meryem_a@hotmail.com)
siham SL lkhoyali (s.lkhoyali@gmail.com)
anouar AB boukir (boukir@yahoo.fr)
mustapha ME elkabous (e-elkabous@gmail.com)
ahlima HA abassain (abahssainhalima@yahoo.fr)
hind HM mrabti (h_mrabti@gmail.com)
basma BE elkhanoussi (dr_basma_2@hotmail.com)
hassan HE errhani (h-errhani@yahoo.fr)

Version: 4
Date: 22 July 2014

Author's response to reviews:

Dr Aitelhaj Meryem 18 July, 2014
Resident on medical oncology
National institute of oncology
m_meryem_a@hotmail.com
TEL: 00212661 16 79 53

Dear Editor

Please find enclosed our manuscript "Breast and splenic metastasis of squamous carcinoma from the uterine cervix: a case report"; for your consideration. We modify the manuscript according to the reviewer comments.

Please find below our responses to the comments:

- We add more details to the case presentation.
- We provide CT image showing the splenic lesion in this version of the article. But we didn’t perform a biopsy because of the absence of this lesion in previous imaging, the history of tumor and the radiological characteristics were suggestive of metastasis.
- We change the title to: Breast and splenic metastases of squamous cell carcinoma from the uterine cervix: a case report
- We correct the tenses and the grammatical errors. (Highlighted in red in the article)

Thank you for your time and consideration;
We look forward to hearing from you, at your earliest convenience

Sincerely, yours
Dr Aitelhaj Meryem